Author's response to reviews

Title: Recurrence of perinatal death in Northern Tanzania: A registry based cohort study

Authors:

Michael J Mahande (michael.mahande@isf.uib.no)
Anne K Dalveit (Anne.Daltveit@isf.uib.no)
Blandina T Mmbaga (blaymt@yahoo.com)
Joseph Obure (jobure2002us@yahoo.com)
Gleard Masenga (drgmasenga@yahoo.com)
Rachel Manongi (rmanongi@yahoo.co.uk)
Rolv T Lie (Rolv.Lie@smis.uib.no)

Version: 6 Date: 9 August 2013

Author's response to reviews: see over
Dear Editor,

Thank you very much for the invitation to resubmit our manuscript entitled “Recurrence of perinatal death in Northern Tanzania: A registry based cohort study” and thanks for the opportunity to further improve the manuscript. All the editorial suggestions have been adopted, and we hope the paper now is acceptable for publication in your journal. Please find our responses indicated in a red colour. Changes in the main text are all in track changes.

Section Editor's Comment:

Few corrections still to be made:
1. Page 3: although one of the reviewers may be suggested to include the word "of", this is not correct: omit "of"
   
   The word “of” has been omitted.

2. Page 8: "for women who lost women who lost their child in the first recorded pregnancy": omit: "who lost women"
   
   We recognize this error, and have corrected the text as suggested.
3. Page 8: add "their" to "for mothers who lost child in first reported pregnancy"

We have added the word “their” as suggested by the editor.

4. Page 13: change "There some possibility that some women ..." in "There is a possibility that some women ..."

This has now been modified as per editor’s suggestion.

5. Page 14: change "... women who deliver a health child or ..." in "... women who deliver a healthy child ...

We have changed to “healthy child”, as correctly pointed out.

6. Page 14: change Similar finding has been reported elsewhere" in "Similar findings have been reported elsewhere"

This has now been corrected.

-------------------------------------------------